Author's response to reviews

Title: Mediastinal abscess after endobronchial ultrasound with transbronchial needle aspiration: a case report

Authors:

Susan D Moffatt-Bruce (susan.moffatt-bruce@osumc.edu)
Patrick Ross Jr. (patrick.ross@osumc.edu)

Version: 2 Date: 14 April 2010

Author's response to reviews:

April 13th, 2010

“Mediastinal abscess after endobronchial ultrasound with transbronchial needle aspiration: a case report”

Dear Sir,

The following changes have been made as requested to the manuscript:

1. The authors list has been written exactly as is and without qualifications.
2. Equal contribution of work has been documented on the title page.
3. Authors contribution has been delineated
4. The title has the proper capitalization
5. The email addresses for each author have been provided.
6. The word count and key words have been removed
7. The abbreviations have been put into a paragraph
8. The reference format has been corrected

I am grateful for the acceptance of this manuscript in this journal.

Sincerely,

Susan Moffatt-Bruce, MD, PhD